Author's response to reviews

Title: Prevalence and distribution of high-risk Human papilloma virus (HPV) types in invasive squamous cell carcinoma of the cervix and in normal women in Andhra Pradesh, India

Authors:
A Pavani Sowjanaya (pavani@cdfd.org.in)
Meenakshi Jain (docmeena@rediffmail.com)
Usha RANI Poli (usharani@eth.net)
Samavedam Padma (padma@cdfd.org.in)
Manik Das (das@yahoo.com)
Keerti V Shah (kvshah@ihsph.edu)
B N Rao (bachin_rao@yahoo.com)
Radha RAMA Devi (radha@cdfd.org.in)
Patti E Gravitt (pgravitt@jhsph.edu)
Gayatri Ramakrishna (gayatri@cdfd.org.in)

Version: 7 Date: 20 December 2005

Author's response to reviews:

To, Dec 20th, 2005

The Editor,
BMC Infectious Disease.

Manuscript #: Manuscript ID 1666681037723737

Sir,

Thanks for accepting our manuscript in the open acess journal of BMC Infectious Disease. We have formatted the manuscript as per your instruction. The following changes have been made:

1) Formatting done as suggested
2) The reference of http://www.canceratlasindia.org has now been deleted from the text and instead a new reference number (23) has been cited both in the text (page 13, line 2) and the Reference list (page 18)

I have formatted the tables as suggested and would like to get clarification if it is agreeable to you. Hope this format is acceptable to you.

Look forward to hear from you.

Thanks

Gayatri Ramakrishna